To The Editor,

Thank you for identifying this problem.

We followed the instructions for correcting problems with reference list extraction. It was found that the field code for one group of references was truncated. To address this issue, each reference was inserted individually and the bibliography was reformatted. Upon resubmission all 71 references were extracted, however the aesthetics of the citations will need to be corrected for the final production.

We hope that this has resolved the issue.

Kind Regards

Rosalind Spence